

GOPEN ACCESS

Citation: Ilboudo K, Boulangé A, Hounyèmè RE, Gimonneau G, Kaboré J, Belem AGM, et al. (2023) Performance of diagnostic tests for *Trypanosoma brucei brucei* in experimentally infected pigs. PLoS Negl Trop Dis 17(11): e0011730. https://doi.org/ 10.1371/journal.pntd.0011730

Editor: Ricardo Toshio Fujiwara, Universidade Federal de Minas Gerais, BRAZIL

Received: May 3, 2023

Accepted: October 16, 2023

Published: November 9, 2023

Copyright: © 2023 Ilboudo et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Data are joint as supplementary information.

Funding: This study has been funded by the Bill & Melinda Gates Foundation in the framework of the Grant Agreement entitled "OPERATIONAL IMPLEMENTATION IN TRANS-BOUNDARY AREAS (INC. PROCUREMENT, SUPPLY CHAIN)" with acronym "Trypa No!" dated September 30th 2016, to VJ, AB, GG, JK, KI and MK. The funder had no role in study design, data collection and analysis, **RESEARCH ARTICLE**

Performance of diagnostic tests for *Trypanosoma brucei brucei* in experimentally infected pigs

Kadidiata Ilboudo^{1,2}, Alain Boulangé^{1,3,4,5}, Robert Eustache Hounyèmè^{1,3}, Geoffrey Gimonneau^{4,5,6}, Jacques Kaboré^{1,2}, Adrien Gaston Marie Belem⁷, Marc Desquesnes^{4,5,8}, Veerle Lejon⁵, Mathurin Koffi⁹, Vincent Jamonneau^{3,5}, Sophie Thévenon^{6,4,5,*}

 Unité de Recherche sur les Maladies à Vecteurs et Biodiversité, Centre International de Recherche-Développement sur l'Elevage en zone Subhumide, Bobo-Dioulasso, Burkina Faso, 2 Unité de Formation et de Recherche en Sciences de la Vie et de la Terre, Université Nazi Boni, Bobo-Dioulasso, Burkina Faso,
Unité de Recherche « Trypanosomoses », Institut Pierre Richet, Bouaké, Côte d'Ivoire, 4 CIRAD, UMR INTERTRYP, Montpellier, France, 5 INTERTRYP, University of Montpellier, CIRAD, IRD, Montpellier, France, 6 Laboratoire National d'Élevage et de Recherches Vétérinaires, Service de Bio-Écologie et Pathologies Parasitaires, Dakar—Hann, Sénégal, 7 Institut du Développement Rural, Université Nazi Boni, Bobo-Dioulasso, Burkina Faso, 8 National Veterinary School of Toulouse (ENVT), Toulouse, France, 9 Laboratoire de Biodiversité et Gestion des Ecosystèmes Tropicaux, Unité de Recherche en Génétique et Épidémiologie Moléculaire, UFR Environnement, Université Jean Lorougnon Guédé, Daloa, Côte d'Ivoire

* sophie.thevenon@cirad.fr

Abstract

Animal African trypanosomosis is an important vector-borne disease of livestock in sub-Saharan Africa. Pigs seem relatively tolerant to trypanosome infection and could act as a reservoir of trypanosomes affecting animals and humans. Our ability to reliably detect trypanosome infection in pigs depends on the performance of diagnostic tools, which is not well known. In pigs experimentally infected with Trypanosoma brucei brucei, we evaluated the performance of parasitological Buffy Coat Technique (BCT), two molecular (TBR and 5.8S PCR) and four serological tests (CATT, HAT Sero-K-Set rapid diagnostic test-RDT, indirect ELISA, immune trypanolysis). Most diagnostic tests showed high specificity, estimated at 100% (95% CI = 74–100%) with the exception of CATT and RDT whose specificity varied between 100% (95% CI = 74-100%) to 50% (95% CI = 7-93%) during the experiment. The sensitivity of each test fluctuated over the course of the infection. The percentage of positive BCT over the infection (30%) was lower than of positive PCR (56% and 62%, depending on primers). Among the serological tests, the percentage of positive tests was 97%, 96%, 86% and 84% for RDT, ELISA, immune trypanolysis and CATT, respectively. Fair agreement was observed between both molecular tests ($\kappa = 0.36$). Among the serological tests, the agreement between the ELISA and the RDT was substantial ($\kappa = 0.65$). Our results on the T.b. brucei infection model suggest that serological techniques are efficient in detecting the chronic phase of infection, PCR is able to detect positive samples several months after parasites inoculation while BCT becomes negative. BCT examination and RDT are useful to get a quick information in the field, and BCT can be used for treatment decision. ELISA appears most suited for epidemiological studies. The selection of

decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

diagnostic tests for trypanosomosis in pigs depends on the context, the objectives and the available resources.

Author summary

The presence of wild and domestic healthy carriers is an obstacle to the control and elimination of animal and human trypanosomosis in Africa. Domestic pigs are receptive to infection by most trypanosomes, and may act as a reservoir of parasites for livestock and humans. Indeed, field studies highlighted that pigs are often positive for different diagnostic tests for trypanosome infections. However, the performances of these tests are poorly known in pigs, while accurate diagnosis of pig trypanosomosis is essential for epidemiological surveillance and implementation of control measures. This study evaluated the diagnostic performance of one parasitological, two molecular and four serological diagnostic tests to detect *Trypanosoma brucei brucei* infection in experimentally infected pigs. The results showed a high specificity of most tests, except for the serological tests CATT and HAT Sero-*K*-Set Rapid Diagnostic Test (RDT). The sensitivity varied highly depending on the tests.

While the choice of a test or a test combination depends on the objectives of a given study and the available resources, parasitological observation can be proposed for quick field screening in conjunction with a RDT, provided that RDT performances are confirmed in a larger study, and indirect antibody-ELISA and PCR are suited for epidemiological investigations.

Introduction

Animal African trypanosomosis (AAT), or nagana, is considered as one of the main health constraints to livestock production in sub-Sahara Africa [1], with an additional indirect impact on crop production [2]. This disease is caused by unicellular protozoan parasites of the genus *Trypanosoma* (order Kinetoplastida). *Trypanosoma brucei brucei*, *T. congolense* and *T. vivax* are the main species responsible for AAT in livestock [3,4]. They are mainly transmitted cyclically by tsetse flies (*Glossina* spp.) [5], but *T. vivax* can also be transmitted mechanically by biting flies such as *Tabanus* spp and *Stomoxys* spp [6,7].

For epidemiological studies, surveillance, planning and implementation of control measures, diagnosis of AAT is based on direct or indirect trypanosomes detection tests [8], since clinical manifestations of AAT can be confused with other parasitic affections or signs of malnutrition, vary by strain and species of the parasite, and also depend on host factors [9]. Direct trypanosome detection by microscopy is commonly considered as the gold standard for confirmation of infection as it is highly specific, though the technique can lack sensitivity [10]. Serological tests based on the detection of specific antibodies, and molecular tests that detect parasite DNA have shown high sensitivities and specificities [10]. The evaluation of these diagnostic tests has so far mainly been carried out on cattle [11,12] that are susceptible to the infection and for which the economic impact of AAT is important [1,2].

Yet, in order to achieve a sustainable control of AAT, it is crucial to consider that different wild and domestic host species may act as reservoir of parasites. The case of local African pigs is particularly challenging since pigs are (i) particularly receptive and generally tolerant to most of the African trypanosomes infections [13–19] and (ii) the preferred host for tsetse,

especially *Glossina palpalis palpalis* [20,21]. In the field, pigs have been found positive for T. brucei in different locations in Africa [14,16-18,22-24]. Trypanosoma brucei brucei is considered less pathogenic than T. congolense and T. vivax in cattle and small ruminants, and T.b. *brucei* infections are often asymptomatic in pigs [3]. However, attention should also be paid to *T.b. brucei*. First, *T.b. brucei* can be responsible for anaemia and severe disease in cattle [25,26], small ruminants [27] and pigs [28,29] leading to death [30]. Second, because it has the ability to leave the blood circulation and invade deep tissues [31,32], T.b. brucei can be difficult to detect by direct blood observation and might escape from trypanocidal drug action [33]. In addition, its differential diagnosis with T.b. rhodesiense and T.b. gambiense, responsible for Human African Trypanosomiasis (HAT) [34] is difficult even though pigs may carry T.b. rhodesiense [22] and are suspected to be a reservoir of T.b. gambiense [35]. In the interest to be able to detect T. brucei s.l., and to distinguish afterwards the different parasite subspecies, the performances of the diagnostic tests must be accurately evaluated. In the field, parasitological observations, PCR and serological tests have been used in pigs to assess both the circulation of animal trypanosomes and of human-infecting trypanosomes, but without a previous knowledge of their performances in terms of sensitivity and specificity [18,36].

In this context, this study aims at evaluating the diagnostic performance of diagnostic tools developed for AAT and HAT to detect T.b. brucei infection in pigs. For this purpose, we carried out an experimental infection of pigs with T.b. brucei and assessed the performances of seven diagnostic tests: parasitological examination based on the Buffy Coat Technique (BCT), two molecular PCR tests and four serological tests: indirect ELISA, immune trypanolysis (TL), Card Agglutination Test for Trypanosomiasis/T.b. gambiense (CATT)—and HAT Sero-K-Set rapid diagnostic test (RDT). The experimental infection as well as the BCT and TL results have already been described in a publication focusing on the T.b. gambiense specificity of TL using LiTat 1.3 and LiTat 1.5 variant antigen types (VAT) [37]. This study suggested that TL using the LiTat 1.6 VAT had sufficient sensitivity to be used as a serological test for the diagnosis of *T.b. brucei* infections in pigs [37]. The CATT and HAT Sero-K-Set are field tests for HAT diagnosis, but since comparable field tests do not exist for T.b. brucei, they were included in the evaluation assuming potential cross-reactivity. In the present study, the performance of the tests was evaluated in terms of sensitivity and specificity. Also taking into account test feasibility, we can conclude that the selection of diagnostic tests for *T.b. brucei* in pigs depends on the context, the objectives and the available resources.

Material and methods

Ethics statement

The study protocol was approved by the *Comité d'Ethique et de Biosécurité* of CIRDES (Centre International de Recherche-Développement sur l'Elevage en zone Subhumide) and the implementing authorisation (070_2020/ADM/DG/nka) was obtained in June 2020. Infected pigs were kept in mosquito net protected stables to avoid contact with potential vectors of trypanosomes. Pigs were observed daily by a veterinarian and any animal showing signs of distress or suffering (weight loss) during the experimental period was euthanized. All animals were euthanized at the end of the experiment.

Experimental infection and sampling

The 6 months experimental infection was already described elsewhere [37]. Briefly, 12 pigs were acquired from a tsetse-free area in western Burkina Faso. Eight were infected with the *T*. *b. brucei* MSUS/CI/2013/BE8P2P2 strain and four constituted the uninfected control group.

Peripheral blood samples for BCT were collected for each animal from the auricular vein in a 70 μ l heparinised capillary tube (75 x 1.5 mm) the day of infection (D0), 7 days post infection (DPI) and twice a week from 7 to 189 DPI. Blood samples were also collected in 5 ml EDTA vacutainer tubes from the central vascular system, 19 days before infection (-19 DPI), 5 days before infection (-5 DPI), the day of infection (D0) and then weekly until 189 DPI. After performing the CATT and the RDT, the remaining blood was centrifuged at 15.000 rpm for 15 min. One ml plasma and 500 μ l buffy coat were collected and stored at -20°C for subsequent serological (indirect ELISA, TL) and molecular (PCR) analyses, respectively.

Diagnostic tests

For parasitological examination, the capillary tube was examined by the BCT [38]. For molecular diagnosis, DNA was processed from the buffy coat samples using the Chelex 100 DNA preparation method [39]. Briefly, 500 µl of buffy coat was mixed with 500 µl of 5% Chelex 100 (Bio-Rad, Foster City, CA, USA) in a 1.5 ml Eppendorf tube. The tube was then incubated at 56°C for 1 h, then at 95°C for 30 min, and centrifuged at 12 000 x g for 3 min, and the supernatant containing DNA was used as a template for PCR. PCR amplification was performed using TBR1/2 (5' - CGAATGAATATTAAACAATGCGCAG-3' / 5' - AGAACCATTTATTAGCTT TGTTGC-3') [40], and 5.8SF/R primers specific of Trypanosomatidae (5'-GCGATGGATGA CTTGGCTTC-3'/5'-TCCCATGCGCCGTTTGCGTTC-3'). These recently designed -in house- 5.8S primers, amplify a short fragment (114bp) of the 5.8S ribosomal DNA and were evaluated here for the first time taking advantage of the experimental conditions. The PCR reactions were carried out in a final volume of 25 µl, composed of molecular biology grade water (16,9 µl), 10X buffer containing 25 mM MgCl2 (2.5 mM final, 2.5 µl), dNTP (deoxyribonucleotide triphosphate) (10 mM, 1 µl), primer (10 mM, 1 µl for each), 0.1 µl of 5 U/µl of FIRE-Pol Taq DNA polymerase (0.5 unit final) and 2.5 µl of template to be tested. The reactions were performed with an initial step of 15 min at 95°c, followed by 35 cycles of 94°C (30 s), 55°C (30 s), and 72°C (30 s) for TBR PCR, and 35 cycles of 94°C (1 min), 50°C (1 min), and 72°C (2 min) for 5.8S PCR, both finishing by an extension at 72°C (5 min). Amplified products were visualized with UV light after electrophoresis in a 2% agarose gel.

Four serological tests were performed. The indirect-ELISA (ELISA) protocol for the detection of antibodies against Trypanosoma was adapted from the World Organisation for Animal Health (founded as OIE) protocol and standardised for the analysis of pig plasma [41,42]. Trypanosoma brucei brucei Farakoba 80/CRTA/1 strain [43] soluble antigens (whole trypanosomes lysate, WTL,) [44] diluted in 50 mM carbonate buffer pH 9.6 at a concentration of 5 µg/ml were used to coat 96-well Polysorp NUNC plates. 100 µl of this dilution was placed in each well and kept overnight at 4°C. The next day, after blocking of the plate for 30 min at 37°C, plasma samples were first diluted 1/20, and then diluted 1/10 during their transfer to the wells with 10 μ l of pre-diluted plasma in 90 μ l of blocking buffer (PBS with 0.1% Tween 20 and 5% skimmed milk) and each sample was deposited in duplicate, as were positive (strongly and weakly) and negative reference samples. After an incubation time of 1 h at 37°C, the plates were emptied and washed three times with washing buffer (PBS with 0.1% Tween 20). Peroxidase conjugate (anti-pig IgG HRPO BIO-RAD) was diluted to 1/20000 in blocking buffer and was added to the well (100 μ l /well) and incubated for 30 minutes. After incubation the plates were emptied, washed 3 times and 100 μ l of ABTS (L-tartaric acid/sodium carbonate) was added to each well. After 30 min at 37°C in the dark, the plates were read at 405 nm wavelength using an ELISA reader. The relative percentage of positivity (RPP) of each sample was calculated from the optical density of the sample in relation to optical densities of positive and negative standards [45]. Any sample

with a RPP less than or equal to 20% was considered negative, while a sample with a RPP higher than 20% was considered positive.

The immune trypanolysis (TL) was performed to test plasma samples as previously described [46]. In the present study, we will only take into account the results obtained using cloned populations of *T.b. gambiense* VAT LiTat 1.6 (TL LiTat 1.6), as that VAT showed the highest sensitivity for the detection of *T.b. brucei* in pigs [37]. Briefly, 25 μ l of plasma was mixed with an equal volume of guinea pig serum, to which 50 μ l of a 10⁷ trypanosomes/ml suspension prepared from infected mouse blood was added. After 90 min of incubation at room temperature, the suspension was examined by microscopy (×250). Trypanolysis was considered positive when 50–100% of the trypanosomes were lysed.

For the CATT [47], twenty-five μ l of EDTA blood was mixed with a drop of CATT reagent on the agglutination card, and spread over the reaction zone. The mixture was then placed on a rotary agitator for 5 min at 60 rounds per minute. The reaction was considered positive in presence of blue agglutinations [47,48].

The HAT Sero-K-Set RDT (Coris BioConcept, Belgium) was used following the manufacturer instructions. Twenty-five μ l of fresh EDTA blood was dispensed in the sample well and two drops of buffer were added. The result was read 15 minutes later.

Statistical analysis and evaluated parameters

Data were entered into Excel and analysed with the R software version 4.0.4 (2021-02-15). The ggplot2 function [49] was used to generate graphs. The specificity and sensitivity of the different tests, and their confidence interval (CI), were assessed with the binomial test. Specificity, (the ability of a test to give a negative result in pathogens-free animals), was determined on the 12 pigs before infection and from the four non-infected ones after infection. Sensitivity, (the ability of a test to give a positive result in infected animals), was assessed over time on the eight infected pigs from 7 DPI. The Friedman test, a non-parametric test for repeated data or block design, and Wilcoxon test for paired data were used to assess the significance of the difference in the date of apparition of the first positive results [50]. Comparison of the sensitivity of the tests was evaluated during the infection using a Generalized Additive Model (gam model) with the R packages mgcv [51,52]. The gam model allows to consider the repeated data at the animal level and to estimate a non-linear temporal effect with smoothing parameters. Parasitological and molecular tests (BCT and both 5.8S and TBR PCR) were compared using the PCR 5.8S as the reference test from 7 DPI, and serological tests (RDT, CATT, ELISA, TL LiTat 1.6) were compared using RDT as the reference test after 7 DPI. Test result (positive or negative) was modelled according to a binomial function (logit link), the diagnostic test effects on the intercept and smoothing factor were assessed, and pig effect was adjusted as random smooths. In addition, agreements between diagnostic tests were assessed by kappa statistic (κ) which quantifies the agreement between two tests for the analysed samples beyond chance and ranges from 0 (no agreement beyond chance) to 1 (perfect agreement beyond chance) using the R package irr [53–55]. Kappa statistics were calculated in the infected group from 7 DPI.

The files containing the raw data and the R script can be found in the <u>S1 Text</u>.

Results

Individual response of the pigs to the parasitological diagnostic test

BCT results have been previously published [37]. Briefly, a total of 642 samples of which 426 came from the infected group (8 pigs) and 216 from the control group (4 pigs) were tested. In the control group, no trypanosomes were detected during the experiment. All pigs in the infected group were trypanosome positive during the experiment. The first positive tests were



Fig 1. Boxplot of the percentages of positive time points to each test estimated per pig. The points represent the percentages of positive tests for each pig of the infected group estimated from 7 DPI until the end of the experiment. The central line represents the median, the box the interquartile range. Each pig is represented by a specific color.

https://doi.org/10.1371/journal.pntd.0011730.g001

observed between 10 and 14 DPI (average 11 ± 2 DPI, median at 10 DPI). The percentage of positive time points in BCT from DPI 7 until the end of the experiment per pig ranged from 19% (95% CI = 9–31% for pig 4) to 47% (95% CI = 31–61% for pig 10) (Fig 1).

Individual result of the pigs in the molecular diagnostic tests

Molecular tests were not performed on the -19 DPI samples and Pig 10 was euthanized at 175 DPI. Both the TBR and the 5.8S PCR were performed on a total of 332 samples of which 220 samples were from the infected group and 112 from the control group. No positive PCR was observed in the control group. Fig 2 shows the individual results of each pig of the infected



Fig 2. Individual result of the eight pigs of the infected group in BCT and PCR tests. Top: positivity in BCT, Bottom left: positivity in 5.8S PCR. Bottom right: positivity in TBR PCR. Large red points: positive test; small red points: negative test; DPI: days post infection.

https://doi.org/10.1371/journal.pntd.0011730.g002

group in the molecular tests. All infected pigs showed positive results for both PCRs. Infection was detected early with 5.8S PCR, all pigs being positive at 7 DPI. With TBR PCR, the first positive tests were detected between 7 and 14 DPI, with a median first date of positivity of 7 DPI and an average of 9±3 DPI. However, during the course of infection, there was a large variability in the percentage of positive time points between pigs, the percentage ranging from 38% (95% CI = 20-59% for pig 1) to 81% (95% CI = 61-93% for pig 4) for TBR, and from 19% (95% CI = 7-39%, for pig 4) to 88% (95% CI = 70-98% for pig 2) for 5.8S PCR (Fig 1). Positive tests were observed until the end of the experiment at 189 DPI, with TBR PCR (pigs 2, 4, 11, 12, 16, 23) and 5.8S PCR (pigs 1, 2, 11, 23).

Individual results of the pigs in the serological tests

The number of analysed samples of the uninfected group was 116 for all tests. For the infected group, there were 225 and 227 plasma samples tested by ELISA and TL LiTat 1.6, respectively, and 228 blood samples tested by CATT and RDT (Pig 10 was euthanized at 175 DPI). The difference in the number of tests was due to an insufficient quantity of collected plasma. The individual result of pigs in the four serological tests is shown in Fig 3. TL LiTat 1.6 results have been previously published [37]. No positive tests were found by ELISA or TL LiTat 1.6 in the control group. However, pig 5 of the control pigs was tested positive in CATT and RDT on a regular basis during the course of the experiment. Pig 17 was positive once with CATT and twice with RDT, while pig 15 was positive twice with RDT only. Only pig 21 remained negative in both CATT and RDT.

All infected animals gave positive results in every test at various time points during infection (Fig 3). The first positivity occurred from 7 to 14 DPI for RDT (average 11±4 DPI, median 14), from 7 to 21 DPI for CATT (average 11±5 DPI, median 7), from 14 to 21 DPI for TL LiTat 1.6 (average 15±2 DPI, median 14), and at 14 DPI for ELISA for all animals.

The percentage of positive time points in ELISA per pig, from 7 DPI, was 96% (95% CI = 80-100) (Fig 1). Individual results observed in RDT were also homogenous, with a percentage of positive time points varying from 91% (95% CI = 71-99% for pig 10) to 100% (95%





https://doi.org/10.1371/journal.pntd.0011730.g003

CI = 87–100% for pigs 2, 11, 12). The percentages of positive time points were more variable between pigs for TL LiTat 1.6 and CATT, from 73% (95% CI = 52–88% for pig 1) to 92% (95% CI = 75–99% for pigs 11, 12) and from 55% (95% CI = 32–76% for pig 10) to 96% (95% CI = 80–100% for pig 2, 4) respectively.

Overall diagnostic performance of the different tests

Dates of first detection of positivity differed significantly between tests (P-value $<10^{-3}$ for the Friedman test), but P-values of paired tests could not be exactly computed due to many ties and zeros. We could just notice a trend of earlier appearance of positivity with molecular tests (PCR 5.8S and TBR) and CATT and RDT, followed by BCT and finally ELISA and TL LiTat 1.6 (Figs 2 and 3).

As no positive samples were detected with BCT, both TBR and 5.8S PCR, ELISA and TL LiTat 1.6 in non-infected pigs, the specificities of these tests were estimated at 100% (95% CI = 74-100%). CATT and RDT displayed positive results in non-infected pigs hence their specificities varied from 100% (95% CI = 74-100%, estimated on 12 pigs before infection) to 50% (95% CI = 7-93%, estimated in the control group) according to the sampling day.

Sensitivities differed greatly between parasitological, molecular and serological techniques. An estimation of the sensitivity of each test throughout infection is illustrated in Fig 4.

The Table 1 presents the specificity and the sensitivity of the diagnostic tests, as well as the percentage of negative and positive tests averaged over the experiment, calculated on non-infected and infected animals respectively. Sensitivity of the BCT varied a lot over the course of the infection, ranging from 0% (95% CI = 0-37%) at 7 DPI and at the end of the experiment, to 100% (95% CI = 63-100%), the percentage of positive tests being 30% over the whole experiment (calculated from 7 DPI). The sensitivity of molecular tests also fluctuated a lot in the course of infection but remained above 0%, and the percentage of positive tests was 56% and 62% for TBR and 5.8S respectively over the whole experiment. After first antibodies detection at 14 DPI, sensitivities remained high and stable for ELISA and RDT, with 96% and 97% of



Fig 4. Evolution of the estimated sensitivity during the experiment for each diagnostic test (up parasitological and PCR tests, down serological tests). DPI: days post-infection. For a minimal sensitivity of 0%, 95% Confidence Interval (CI) = 0-37%, for a maximal sensitivity of 100%, 95% CI = 63-100%, for a medium sensitivity of 50%, 95% CI = 16-84%.

https://doi.org/10.1371/journal.pntd.0011730.g004

Table 1. Specificity and sensitivity of different diagnostic tests. Specificity and the percentage of negative tests were estimated during the experiment in non-infected animals (all animals before infection and in the control group), the first one was assessed overtime and the last one averaged on all the samples. Sensitivity and the percentage of positive tests were estimated during the experiment in the infected group from 7 DPI, the first one was assessed overtime and the last one averaged on all the samples. Regarding specificity and sensitivity, minimal and maximal values observed during the experiment are shown on the first and second lines respectively, if they are different. 95% confidence intervals are indicated in brackets.

| Tests | Specificity | Percentage of negative tests | Sensitivity | Percentage of positive tests |
|-------------|-------------------------------|------------------------------|-------------------------------|------------------------------|
| ВСТ | 100% (74–100%) | 100% (98–100%) | 0% (0-37%) 100% (63-100%) | 30% (26–35%) |
| TBR PCR | 100% (74–100%) | 100% (97–100%) | 14% (0–58%) 100% (63–100%) | 56% (43-63%) |
| 5.8S PCR | 100% (74–100%) | 100% (97–100%) | 14% (0-58%) 100% (63-100%) | 62% (55–68%) |
| RDT | 50% (7–93%) 100% (74–100%) | 81% (74–87%) | 0% (0-37%) 100% (63-100%) | 97% (93–98%) |
| ELISA | 100% (74–100%) | 100% (97–100%) | 0% (0-37%) 100% (63-100%) | 96% (92–98%) |
| CATT | 50% (7–93%) 100% (74–100%) | 83% (76-87%) | 50% (16-84%) 100% (63-00%) | 84% (78-89%) |
| TL LiTat1.6 | 100% (74–100%) | 100% (97–100%) | 0% (0-37%) 100% (63-100%) | 86% (80–90%) |

https://doi.org/10.1371/journal.pntd.0011730.t001

positive tests respectively. The sensitivities of CATT and TL LiTat 1.6 were 50% (95% CI = 16-84%) to 100% (95% CI = 63-100%) from 14 DPI, the percentage of positive tests being 84% and 86% for CATT and TL LiTat 1.6 respectively from 7 DPI.

Sensitivity was first compared for the parasitological and molecular tests using the gam model. TBR PCR did not significantly differ in term of sensitivity from the 5.8S PCR (P-value = 0.13 on the intercept and P-value = 0.80 on the smoothing terms using the gam model). On the contrary, the BCT sensitivity was significantly lower than the one of 5.8S PCR (estimate = -1.98, P-value = 10^{-14} on the intercept and P-value = 10^{-15} on the smoothing terms).

For serological tests, ELISA did not significantly differ in sensitivity from RDT (P-value = 0.99 on the intercept and on the smoothing terms using the gam model), while CATT and TL LiTat 1.6 showed a significant lower sensitivity than RDT (effect = -2.87, P-value = 10^{-4} on the intercept for CATT, and effect = -2.45, P-value = 10^{-3} on the intercept for TL LiTat 1.6).

Concordance between the different diagnostic tests

The data on the agreement between diagnostic tests are presented in Table 2. Agreement was estimated on the one hand for BCT and PCR tests. It was poor between the two PCR tests and the BCT ($0 < \kappa < 0.20$). Fair agreement was observed between the two PCR tests ($\kappa = 0.36$). On the other hand, among the serological tests, the agreement was poor between ELISA and CATT, CATT and TL LiTat 1.6, CATT and RDT ($0 < \kappa < 0.20$). Conversely, there was moderate agreement between the ELISA and the TL LiTat 1.6 ($\kappa = 0.41$) and substantial agreement between the ELISA and the RDT ($\kappa = 0.65$).

Discussion

The diagnosis of trypanosomoses is a basic requirement for epidemiological studies as well as for the implementation and follow-up of control measures. Compared to cattle, pigs are often an overlooked species since they usually express marginal symptoms of trypanosomosis [13,56]. However, paying attention to pigs is essential for AAT control and sustainable elimination of HAT, as pigs may act as a potential reservoir of both animal and human

Table 2. Agreement between molecular and parasitological tests on the one hand, and between serological diagnostic tests on the other hand, estimated in the infected group. Kappa range interpretation: <0 = Disagreement, 0-0.2 = poor agreement, 0.21-0.4 = fair agreement, 0.41-0.6 = moderate agreement, 0.61-0.8 = substantial agreement, 0.81-1 = almost perfect agreement.

| Pair of compared tests | Карра (к) | Interpretation |
|------------------------|-----------|----------------|
| TBR PCR-BCT | 0.004 | Poor |
| 5.8S PCR-BCT | 0.10 | Poor |
| TBR PCR-5.8S PCR | 0.36 | Fair |
| ELISA-RDT | 0.65 | Substantial |
| ELISA-CATT | 0.08 | Poor |
| ELISA-TL LiTat 1.6 | 0.41 | Moderate |
| CATT-TL LiTat 1.6 | 0.05 | Poor |
| CATT-RDT | 0.10 | Poor |
| RDT-TL LiTat 1.6 | 0.35 | Fair |
| ELISA-TBR PCR | -0.04 | Disagreement |
| CATT-TBR PCR | -0.10 | Disagreement |
| RDT-TBR PCR | -0.05 | Disagreement |
| TL LiTat 1.6-TBR PCR | -0.04 | Disagreement |
| ELISA-5.8S PCR | -0.08 | Disagreement |
| CATT-5.8S PCR | -0.23 | Disagreement |
| RDT-5.8S PCR | -0.07 | Disagreement |
| TL LiTat 1.6–5.8S PCR | -0.10 | Disagreement |

https://doi.org/10.1371/journal.pntd.0011730.t002

trypanosomes [35]. The prevalence of trypanosomes in pigs living in tsetse-infested areas should therefore be monitored as accurately as possible, which requires efficient diagnostic tools. Our study is the first to evaluate the diagnostic performance of seven tests developed for AAT or HAT including one parasitological, two molecular and four serological tests, in pigs experimentally infected with *T.b. brucei*.

All tests showed high specificity, reaching up to 100% (95% CI = 74–100%), with the exception of CATT and RDT whose specificity was assessed between 50% (95% CI = 7–93%) to 100% (95% CI = 74–100%) due to some positive samples in non-infected pigs, especially in one pig which was positive to CATT and RDT for most of the experiment while remaining negative to BCT, both PCR, ELISA and TL (including other VATs LiTat 1.3 and LiTat 1.5 [37]). These false positives are likely attributed to cross-reactions with antibodies targeting other pathogens [57]. The SD Bioline HAT RDT [58], another serological test initially developed for HAT diagnosis, also showed a low specificity when used in animals [59].

Furthermore, CATT and HAT Sero-*K*-Set RDT developed for HAT diagnosis are based on variant antigens of *T.b. gambiense* [60,61]. In humans, their specificity ranges from 88% to 99% depending on the study [47,48,60,62,63]. The use of diluted plasma instead of whole blood could improve specificity, at the expense of sensitivity, as it is proposed for humans [64]. We chose to use whole blood in the present study, implementing the CATT as a simple field test for antibody detection.

We also compared the dates at which tests became positive for the first time, which defines the prepatent period. The infection was detected the earliest by molecular tests at a median of 7 DPI for 5.8S PCR and TBR PCR, respectively, as compared to parasitological detection by BCT occurring between 10 and 14 DPI (median at 10 DPI). Early detection of infection by PCR were reported in experimental studies using different trypanosomes and hosts species [12,65,66]. Specific antibodies were first detected by CATT (7–21 DPI) and RDT (7–14 DPI), followed by ELISA (14 DPI) and TL LiTat 1.6 (14–21 DPI). The delay of detection in serology

was expected according to the delay of production of specific antibodies and based on results obtained in experimental infection with *T. vivax* and *T. evansi* [12,67].

The sensitivity of the tests was also assessed. First, the BCT, which is amongst the most sensitive parasitological tests [68,69], was compared to molecular tests (TBR and 5.8S PCR). The percentage of positive tests with BCT (30%) was lower than with TBR PCR (56%) and 5.8S PCR (61%), and BCT was significantly less sensitive than molecular tests. PCR has been reported to be more sensitive than parasitological techniques in experimental conditions [12,65,66,70,71] and when applied to field samples [16,72,73]. We also observed that the BCT sensitivity decreased from 30 DPI onward. BCT false negative results are common when parasitaemia is low in peripheral blood, below the detection threshold [11], and it has a low sensitivity in the field [24,74]. Sensitivities of PCRs were also highly fluctuating, though always higher than BCT. Interestingly, while all pigs were negative to BCT during the last month of the experiment, molecular tests kept giving positive results, highlighting the persistence of the trypanosomal DNA in the bloodstream. A higher sensitivity of the TBR PCR versus 5.8S PCR was expected, on the basis of previous studies that compared amplification of the satellite DNA, thought to be present in 10,000 to 20,000 copies per genome [75–77], and ribosomal DNA [70,78], repeated around 100–200 times per genome [77,79]. Difference in primers affinity for the targeted sequences could explain this observation. Alternatively, diversity in the sequence targeted by TBR primers could explain a lack of sensitivity of the TBR PCR, as suggested by Van Reet et al. (2021) [80]. Performances of the new 5.8S PCR will deserve complementary analyses to accurately assess its analytical sensitivity and specificity in different types of samples, coming from various host species and infection types.

When comparing sensitivity of antibody detection tests, it appeared that all serological tests successfully detected *T.b. brucei* infection in pigs. The responses were more stable for ELISA and RDT with 96% and 97% of positive tests respectively, when calculated from 7 DPI. However, from 14 DPI, the sensitivity of ELISA was 100% until the end of the experiment. Overall, CATT and TL LiTat 1.6 showed lower sensitivities than did RDT and ELISA, and their sensitivity also fluctuated during the course of the experiment. The ELISA antibody-detection test developed here used whole soluble antigens of *T.b. brucei*. Such a test making use of WTL has generally high sensitivity [12] as it provides numerous antigen targets that can be recognized by antibodies. The fluctuating sensitivity of the CATT during the chronic phase could be due to the fact that CATT detects mainly IgM. For example, in cattle trypanosomosis due to *T. congolense*, IgM rates were lower than IgG rates. As previously shown [37], sensitivity of TL LiTat 1.6 for *T.b. brucei* infection in pigs was high, but it seemed less than for surveillance of gambiense-HAT [46,81], for which TL was developed. Overall, higher sensitivity of serological tests than of BCT and molecular tests was expected and can be explained by the sustained production and persistence of specific antibodies, whereas the parasitaemia is fluctuating [82,83].

Agreements among the methods varied from poor (no agreement between positive and negative results of a pair of tests for a same sample) to substantial (agreement between positive and negative results). Agreement between molecular tests and BCT was poor. That some samples were positive to one or two molecular tests and negative to BCT could be expected from the difference in their sensitivities [11]. However, some parasitology-positive samples were negative by PCR. Parasitological examinations were done on peripheral blood collected from the auricular vein, while PCR was performed on blood collected from the anterior vena cava. Local concentration of parasites could differ depending on blood origin explaining this result. More surprising was the fair agreement between both PCR. These unexpected results may be due to technical failures related to the DNA preparation method or amplification inhibiting factors [84]. ELISA and RDT showed a high agreement while CATT provided low agreement with other serological tests, maybe suggesting fluctuating IgM concentrations. As expected,

Kappa indexes calculated between serological and molecular tests were negative, highlighting disagreement between the two types tests that identify totally different elements, host antibodies and parasite DNA respectively that present different dynamics, but that can thus be viewed as complementary.

Given the relatively small number of animals, as well as the individual variations between them, the observed indicators of performance had large confidence intervals and should be interpreted with caution. The use of experimental infections may somehow bias the estimated diagnostic sensitivity and specificity compared to natural infections [85]. Indeed, a better specificity for all tests can be expected for experimental infections, as pigs were initially treated against various infections, while in the field animals remain exposed to many pathogens, including other trypanosome species. Sensitivity could be also biased in experimental infection due to the use of a specific trypanosome strain, the inoculation dose and the administration route, that differ from natural infections. Furthermore, it must be noted that, to precisely assess the prepatent period, samples should have been collected more frequently during the first three weeks of the experiment, which could not be performed due to the stress caused to pigs.

Despite these limitations, we can provide some conclusions on tests performance and recommendations to the potential use of these diagnostic tests to assess pig infection by T.b. brucei. High specificities were recorded for BCT, both PCRs, ELISA and TL LiTat1.6, but observed specificities were lower for RDT and CATT. On the other hand, ELISA and TDR showed high sensitivities, followed by TL LiTat1.6 and CATT. The PCRs displayed a moderate sensitivity, and BCT a poor one, especially in late phase of infection. Serological techniques are efficient in detecting infected animals, especially during the chronic phase of infection, but it must be reminded that a positive test does not necessarily reflects active infections since antibodies may persist after treatment or self-cure [86,87]. PCRs were able to detect positive samples until six months post-inoculation while parasitaemia had become undetectable, signalling the persistent presence of trypanosomes. The joint use of both PCRs (TBR and 5.8S), run in parallel, strengthened the global sensitivity of our molecular tests, the cumulated percentage of positive tests reaching 75%, thus decreasing the global risk of getting false-negative results without deteriorating the specificity. In terms of practical use, BCT, TDR and CATT can be performed in the field. TDR is a so-called point of care test, CATT is suitable to quickly screen dozens of animals, while BCT requires only basic equipment, a centrifuge and a microscope, and is easy to perform by a trained staff. ELISA, PCR and TL require a well-equipped laboratory and well-trained technicians. In addition, TL needs animal facilities to multiply live parasites and must therefore be performed in a reference laboratory [88].

In conclusion, the selection of tests to be used to detect *T.b. brucei* in pigs will depend on the context, the objectives and the financial and technical resources available according to the elements mentioned above. Parasitological detection by BCT and the use of RDT and/or CATT can be promoted in the field to identify early infections and support treatment decision for BCT, and to have an immediate impression of the epidemiological situation for CATT and RDT. ELISA appears to be the most reliable and suitable laboratory serological test for epidemiological surveys and for the monitoring of control campaigns during which prevalence is expected to decrease if control is efficient. The use of PCR is also important to identify ongoing infections and is adequate to assess the presence of animal reservoirs of trypanosomes. For that purpose, the use of two pairs of primers targeting different DNA sequences could be proposed. It should be kept in mind that performances of the tests for other trypanosome species infecting pigs have not been assessed here and remain to be compared. Furthermore, as our result suggest that CATT and RDT lack specificity, additional research is needed to develop point-of-care diagnostics that should be accurate, cheap, and easy to use [89].

Supporting information

S1 Table. diagnostic_BCT.csv. Raw data for buffy coat test. Columns header: Index (index line), Date: date of sampling; Pig: pig identifier; Group: group C for the control group, I for infected group; Result1: result of the BCT test 0 for negative, 1 for positive; DPI: Date Post-Infection; Phase: pre_inf for before infection; inf for from infection; Status: healthy or infected status of pigs; PA.ml: log10 of parasitemia in trypanosomes/ml. (CSV)

S2 Table. diagnostic_PCR.csv. Raw data for PCR tests. Columns header: Index (index line), Date: date of sampling; Group: group C for the control group, I for infected group; Pig: pig identifier; Test: molecular test (TBR PCR, 5.8S PCR); Result: result of the PCR test 0 for negative, 1 for positive; DPI: Date Post-Infection; Phase: pre_inf for before infection; inf for from infection; Status: healthy or infected status of pigs. (CSV)

S3 Table. diagnostic_Serologic.csv. Raw data for serological tests. Columns header: Index (index line), Date: date of sampling; Group: group C for the control group, I for infected group; Pig: pig identifier; Test: serolgoical test (CATT, RDT, ELISA, TL LiTat1.6); Result: result of the serological test 0 for negative, 1 for positive; DPI: Date Post-Infection; Phase: pre_-inf for before infection; inf for from infection; Status: healthy or infected status of pigs. (CSV)

S4 Table. Prepatente_all_diag.csv. Table containing first date of positivity for each pig and each diagnostic tests. Pig: pig identifier; Test: diagnostic tests (BCT, TBR PCR, 5.8S PCR, CATT, RDT,ELISA, TL LiTat1.6); PP: time point of first positivity in DPI; Group: C for the control group, I for infected group. (CSV)

S1 Text. Script_diagnostic_tools.R. R script for statistical analyses. (R)

Acknowledgments

We acknowledge all the technicians of *Centre International de Recherche-Développement sur l'Elevage en zone Subhumide (CIRDES*, Bobo-Dioulasso, Burkina-Faso) that contributed to this experimental infection. We thank Jean-Mathieu Bart for his fruitful advices regarding the manuscript.

Author Contributions

Conceptualization: Alain Boulangé, Geoffrey Gimonneau, Jacques Kaboré, Adrien Gaston Marie Belem, Marc Desquesnes, Mathurin Koffi, Vincent Jamonneau.

Data curation: Kadidiata Ilboudo, Sophie Thévenon.

Formal analysis: Kadidiata Ilboudo, Sophie Thévenon.

Funding acquisition: Jacques Kaboré, Mathurin Koffi, Vincent Jamonneau.

Investigation: Kadidiata Ilboudo, Alain Boulangé, Robert Eustache Hounyèmè, Geoffrey Gimonneau, Jacques Kaboré.

Methodology: Kadidiata Ilboudo, Alain Boulangé, Geoffrey Gimonneau, Jacques Kaboré, Vincent Jamonneau, Sophie Thévenon.

- **Project administration:** Alain Boulangé, Geoffrey Gimonneau, Jacques Kaboré, Vincent Jamonneau.
- **Resources:** Alain Boulangé, Robert Eustache Hounyèmè, Geoffrey Gimonneau, Jacques Kaboré, Vincent Jamonneau.
- Software: Sophie Thévenon.
- Supervision: Jacques Kaboré, Adrien Gaston Marie Belem, Marc Desquesnes, Veerle Lejon, Mathurin Koffi, Vincent Jamonneau.
- Validation: Alain Boulangé, Sophie Thévenon.
- Visualization: Kadidiata Ilboudo, Vincent Jamonneau, Sophie Thévenon.
- Writing original draft: Kadidiata Ilboudo, Alain Boulangé, Geoffrey Gimonneau, Veerle Lejon, Vincent Jamonneau, Sophie Thévenon.
- Writing review & editing: Kadidiata Ilboudo, Alain Boulangé, Robert Eustache Hounyèmè, Geoffrey Gimonneau, Jacques Kaboré, Adrien Gaston Marie Belem, Marc Desquesnes, Veerle Lejon, Mathurin Koffi, Vincent Jamonneau, Sophie Thévenon.

References

- Kristjanson PM, Swallow BM, Rowlands GJ, Kruska RL, de Leeuw PN. Measuring the costs of African animal trypanosomosis, the potential benefits of control and returns to research. Agric Syst. 1999; 59: 79–98.
- 2. Swallow BM. Impacts of trypanosomiasis on African agriculture. Rome: FAO. 2000; 46 p.
- 3. Taylor K, Authié EML. Pathogenesis of animal trypanosomiasis. In: Maudlin I., Holmes P. H. and Miles M. A. The Trypanosomiases. Wallingford, UK; 2004. pp. 331–353.
- Uilenberg G, Boyt WP. A field guide for the diagnosis, treatment and prevention of African animal trypanosomosis. New ed. Rome: Food and Agriculture Organization of the United Nations; 1998; https://www.fao.org/3/x0413/e00.htm
- Jordan AM. Tsetse flies as vectors of trypanosomes. Vet Parasitol. 1976; 2: 143–152. <u>https://doi.org/10.1016/0304-4017(76)90059-5</u>
- 6. Desquesnes M, Dia ML, Acapovi G, Yoni W. Les vecteurs mécaniques des trypanosomoses animales: généralité, morphologie, biologie, impacts et contrôle. Identification des espèces les plus abondantes en Afrique de l'Ouest. CIRDES. 2005; 68 p.
- Desquesnes M, Dia ML. Mechanical transmission of *Trypanosoma vivax* in cattle by the African tabanid Atylotus fuscipes. Vet Parasitol. 2004; 119: 9–19. <u>https://doi.org/10.1016/j.vetpar.2003.10.015</u> PMID: 15036572
- Diall O, Cecchi G, Wanda G, Argilés-Herrero R, Vreysen MJB, Cattoli G, et al. Developing a Progressive Control Pathway for African Animal Trypanosomosis. Trends Parasitol. 2017; 33: 499–509. https://doi.org/10.1016/j.pt.2017.02.005 PMID: 28456474
- Courtin D, Berthier D, Thevenon S, Dayo G-K, Garcia A, Bucheton B. Host genetics in African trypanosomiasis. Infect Genet Evol. 2008; 8: 229–238. https://doi.org/10.1016/j.meegid.2008.02.007 PMID: 18394971
- Desquesnes M, Gonzatti M, Sazmand A, Thévenon S, Bossard G, Boulangé A, et al. A review on the diagnosis of animal trypanosomoses. Parasit Vectors. 2022; 15: 64. https://doi.org/10.1186/s13071-022-05190-1 PMID: 35183235
- Alves W, Cuglovici Abrão D, Fernando L, Furtado V, Silveira J, Flávio M, et al. Comparison of three methods for diagnosis of *Trypanosoma* (Duttonella) *vivax* in cattle. Genetics and Molecular Research. 2017; 16: gmr16039811. http://dx.doi.org/10.4238/gmr16039811
- Fidelis Junior OL, Sampaio PH, Gonçalves LR, André MR, Machado RZ, Wijffels G, et al. Comparison of conventional and molecular techniques for *Trypanosoma vivax* diagnosis in experimentally infected cattle. Braz. J. Vet. Parasitol. 2019; 28: 203–209. <u>https://doi.org/10.1590/s1984-29612019034</u> PMID: 31188948
- **13.** Deckers N. Experimental infection of *trypanosoma congolense* in pigs and a contribution to the improved diagnosis of trypanosomosis in pigs. M.Sc. Thesis. ITMA–MSCTAH. 2004; 29 p.

- Karshima SN, Ajogi I, Mohammed G. Eco-epidemiology of porcine trypanosomosis in Karim Lamido, Nigeria: prevalence, seasonal distribution, tsetse density and infection rates. Parasit Vectors. 2016; 9: 448. https://doi.org/10.1186/s13071-016-1732-x PMID: 27519526
- Mehlitz D. Le réservoir animal de la maladie du sommeil à *Trypanosoma brucei gambiense*. Etudes et Synthèses de l'IEMVT. n.18, Maisons-Alfort: CIRAD-IEMVT. 1986,156 p. ISBN 2-85985-127-5
- 16. N'Djetchi MK, Ilboudo H, Koffi M, Kaboré J, Kaboré JW, Kaba D, et al. The study of trypanosome species circulating in domestic animals in two human African trypanosomiasis foci of Côte d'Ivoire identifies pigs and cattle as potential reservoirs of *Trypanosoma brucei gambiense*. PLoS Negl Trop Dis. 2017; 11: e0005993. https://doi.org/10.1371/journal.pntd.0005993 PMID: 29045405 PMCID: PMC5662240
- Njiokou F, Nimpaye H, Simo G, Njitchouang GR, Asonganyi T, Cuny G, et al. Domestic animals as potential reservoir hosts of *Trypanosoma brucei gambiense* in sleeping sickness foci in Cameroon. Parasite. 2010; 17: 61–66. https://doi.org/10.1051/parasite/2010171061 PMID: 20387740
- Traoré BM, Koffi M, N'Djetchi MK, Kaba D, Kaboré J, Ilboudo H, et al. Free-ranging pigs identified as a multi-reservoir of *Trypanosoma brucei* and *Trypanosoma congolense* in the Vavoua area, a historical sleeping sickness focus of Côte d'Ivoire. PLoS Negl Trop Dis. 2021; 15: e0010036. https://doi.org/10. 1371/journal.pntd.0010036 PMID: 34937054 PMCID: PMC8735613
- Vourchakbé J, Tiofack AAZ, Mbida M, Simo G. Trypanosome infections in naturally infected horses and donkeys of three active sleeping sickness foci in the south of Chad. Parasit Vectors. 2020; 13: 323. https://doi.org/10.1186/s13071-020-04192-1 PMID: 32576240 PMCID: PMC7310289
- Dagnogo M, Traoré G, Koné M. Évaluation de l'attractivité de *Glossina palpalis palpalis* vis-à-vis de l'homme et du porc dans la région de Daloa en Côte d'Ivoire. Revue Élev Méd Vét Pays Trop. 1996; 49: 50–54. https://doi.org/10.19182/remvt.9546
- Sané B, Laveissière C, Méda HA. Répartition spatiale et préférences trophiques de Glossina palpalis palpalis dans le foyer forestier de Zoukougbeu (Cote d'Ivoire). Implications épidémiologiques. Parasite. 2000; 7: 241–244. https://doi.org/10.1051/parasite/2000073241 PMID: 24499540 PMCID: PMC3843548
- Hamill LC, Kaare MT, Welburn SC, Picozzi K. Domestic pigs as potential reservoirs of human and animal trypanosomiasis in Northern Tanzania. Parasit Vectors. 2013; 6: 322. <u>https://doi.org/10.1186/</u> 1756-3305-6-322 PMID: 24499540 PMCID: PMC3843548
- Waiswa C. Porcine trypanosomiasis in southeastern Uganda: prevalence and assessment of therapeutic effectiveness. Bulg. J. Vet. Med. 2005; 8:59–68.
- Ebhodaghe F, Ohiolei JA, Isaac C. A systematic review and meta-analysis of small ruminant and porcine trypanosomiasis prevalence in sub-Saharan Africa (1986 to 2018). Acta Tropica. 2018; 188: 118– 134. https://doi.org/10.1016/j.actatropica.2018.08.034 PMID: 30179607
- Doko A, Verhulst A, Pandey V, Stuyft P. Trypanosomose expérimentale à *Trypanosoma brucei brucei* chez les taurins Holstein et les zébus Bororo blancs. Revue Élev Méd vét Pays trop. 1997; 49:207– 211. https://doi.org/10.19182/remvt.9597
- Dargie JD, Murray PK, Murray M, McIntyre WIM. The blood volumes and erythrokinetics of Ndama and Zebu cattle experimentally infected with *Trypanosoma brucei*. Res Vet Sci. 1979; 26: 245–247. https:// doi.org/10.1016/S0034-5288(18)32926-6 PMID: 262609
- Geerts S, Osaer S, Goossens B, Faye D. Trypanotolerance in small ruminants of sub-Saharan Africa. Trends Parasitol. 2009; 25: 132–138. https://doi.org/10.1016/j.pt.2008.12.004 PMID: 19200783
- Omeke BCO, Ugwu DO. La trypanosomose porcine: anémie et histologie comparées des organes lymphoïdes. Revue Élev Méd Vét Pays Trop. 1991; 44: 267–272. https://doi.org/10.19182/remvt.9164
- Atata JA, Enam SJ, Ogbuagu NE, Balogun EO, Adamu S, Esievo KAN. Upregulation of sialyltransferases ST3Gal1 and ST6Gal1 promotes stabilization of erythrocyte mass and recovery of anemia in *Trypanosoma brucei*-infected pigs. Res Vet Sci. 2022; 145: 102–108. <u>https://doi.org/10.1016/j.</u> rvsc.2022.02.012 PMID: 35180660 PMCID: PMC9148701
- Agu WE, Bajeh ZT. An outbreak of fatal *Trypanosoma brucei* infection of pigs in Benue State of Nigeria. Trop Vet. 1986; 4:25–25
- Kangethe RT, Boulangé A, Coustou V, Baltz T, Goetzer THT. *Trypanosoma brucei brucei* oligopeptidase B null mutants display increased prolyl oligopeptidase-like activity. Mol Biochem Parasitol. 2012; 182:7–16. https://doi.org/10.1016/j.molbiopara.2011.11.007 PMID: 22123425
- Trindade S, Rijo-Ferreira F, Carvalho T, Pinto-Neves D, Guegan F, Aresta-Branco F, et al. *Trypanosoma brucei* Parasites Occupy and Functionally Adapt to the Adipose Tissue in Mice. Cell Host Microbe. 2016; 19: 837–848. https://doi.org/10.1016/j.chom.2016.05.002 PMID: 27237364 PMCID: PMC4906371
- Claes F, Vodnala SK, Reet N van, Boucher N, Lunden-Miguel H, Baltz T, et al. Bioluminescent Imaging of *Trypanosoma brucei* Shows Preferential Testis Dissemination Which May Hamper Drug Efficacy in

Sleeping Sickness. PLoS Negl Trop Dis. 2009; 3: e486. https://doi.org/10.1371/journal.pntd.0000486 PMID: <u>19621071</u> PMCID: PMC2707598

- Büscher P, Cecchi G, Jamonneau V, Priotto G. Human African trypanosomiasis. The Lancet. 2017; 390: 2397–2409. https://doi.org/10.1016/S0140-6736(17)31510-6 PMID: 28673422
- 35. Büscher P, Bart J-M, Boelaert M, Bucheton B, Cecchi G, Chitnis N, et al. Do Cryptic Reservoirs Threaten Gambiense-Sleeping Sickness Elimination? Trends Parasitol. 2018; 34: 197–207. <u>https://doi.org/10.1016/j.pt.2017.11.008</u> PMID: 29396200 PMCID: PMC5840517
- Karshima SN, Lawal IA, Bata SI, Barde IJ, Adamu PV, Salihu A, et al. Animal Reservoirs of *Trypano-soma Brucei Gambiense* around the Old Gboko Sleeping Sickness Focus in Nigeria. J. Parasitol. Vector Biol. 2016; 8: 47–54. https://doi.org/10.5897/JPVB2015.0228 PMID: 26941995 PMCID: PMC4752997
- Ilboudo K, Hounyeme RE, Kaboré J, Boulangé A, Gimonneau G, Salou EW, et al. Experimental evidence that immune trypanolysis using the LiTat 1.3 and LiTat 1.5 variant antigen types is not specific to *Trypanosoma brucei gambiense* in pigs. Parasite. 2022; 29:61. <u>https://doi.org/10.1051/parasite/</u>2022063 PMID: 36562442 PMCID: PMC9879134
- **38.** Murray M, Murray PK, McIntyre WI. An improved parasitological technique for the diagnosis of African trypanosomiasis. Trans R Soc Trop Med Hyg. 1977; 71: 325–326. <u>https://doi.org/10.1016/0035-9203</u> (77)90110-9 PMID: 563634
- Walsh PS, Metzger DA, Higuchi R. Chelex 100 as a medium for simple extraction of DNA for PCRbased typing from forensic material. BioTechniques. 2013; 54: 134–139. <u>https://doi.org/10.2144/</u> 000114018 PMID: 1867860
- Moser DR, Cook GA, Ochs DE, Bailey CP, McKane MR, Donelson JE. Detection of Trypanosoma congolense and Trypanosoma brucei subspecies by DNA amplification using the polymerase chain reaction. Parasitology. 1989; 99: 57–66. https://doi.org/10.1017/s0031182000061023 PMID: 2797872
- Luckins AG. Detection of antibodies in Trypanosome-infected cattle by means of a microplate enzymelinked immunosorbent assay. Trop Anim Health Prod. 1977; 9: 53–62. <u>https://doi.org/10.1007/</u> BF02297393 PMID: 333677
- 42. OIE. Nagana: Infections with salivarian trypanosomoses (excluding *Trypanosoma evansi* and *T. equiperdum*) Chapter 3.4.14. Terrestrial Manual. 2021; 18 p.
- Duvallet G, Bengaly Z, Reifenberg J-M, Argiro L. De nouveaux outils pour le diagnostic et l'épidémiologie de la trypanosomose animale africaine. Edition AUPELF-UREF. Paris, pp. 19–29.
- Engvall E, Perlmann P. Enzyme-linked immunosorbent assay (ELISA) quantitative assay of immunoglobulin G. 1971; 8: 871–874. https://doi.org/10.1016/0019-2791(71)90454-x PMID: 5135623
- 45. Desquesnes M, Michel J, Rocque SDL, Solano P, Millogo L, Bengaly Z, et al. Enquête parasitologique et sérologique (Elisa-indirect) sur les trypanosomoses des bovins dans la zone de Sidéradougou, Burkina Faso. Revue Élev. Méd. vét. Pays trop. 1999; 52: 223–232
- 46. Jamonneau V, Bucheton B, Kaboré J, Ilboudo H, Camara O, Courtin F, et al. Revisiting the Immune Trypanolysis Test to Optimise Epidemiological Surveillance and Control of Sleeping Sickness in West Africa. PLoS Negl Trop Dis. 2010; 4: e917. <u>https://doi.org/10.1371/journal.pntd.0000917</u> PMID: 21200417 PMCID: PMC3006129
- Magnus E, Vervoort T, Van Meirvenne N. A card-agglutination test with stained trypanosomes (C. A. T. T.) for the serological diagnosis of *T. b. gambiense* tripanosomiasis. Ann Soc Belg Med Trop. 1978; 58: 169–176. PMID: 747425
- Jamonneau V, Camara O, Ilboudo H, Peylhard M, Koffi M, Sakande H, et al. Accuracy of Individual Rapid Tests for Serodiagnosis of *Gambiense* Sleeping Sickness in West Africa. PLoS Negl Trop Dis. 2015; 9: e0003480. https://doi.org/10.1371/journal.pntd.0003480 PMID: 25642701 PMCID: PMC4314185
- Wickham H. ggplot2: Elegant Graphics for Data Analysis. Springer-Verlag New York; 2016. Available from: https://ggplot2.tidyverse.org.
- 50. Hollander M, Douglas AW. Nonparametric Statistical Methods. John Wiley&Sons. New York; 1973.
- Wood SN. Generalized Additive Models: An Introduction with R. Chapman and Hall/CRC; 2017. https:// doi.org/10.1201/9781315370279
- Wood S. Fast stable restricted maximum likelihood and marginal likelihood estimation of semiparametric generalized linear models. Journal of the Royal Statistical Society (B). 2011; 73: 3–36.
- Cohen J. Weighted kappa: Nominal scale agreement provision for scaled disagreement or partial credit. Psychol Bull. 1968; 70: 213–220. https://doi.org/10.1037/h0026256 PMID: 19673146
- 54. Fleiss JL, Cohen J, Everitt BS. Large sample standard errors of kappa and weighted kappa. Psychol Bull. 1969; 72: 323–327. https://doi.org/10.1037/h0028106

- Gamer M, Jim L, Ian FPS. irr: Various Coefficients of Interrater Reliability and Agreement. 2019. Available from: https://CRAN.R-project.org/package=irr.
- 56. Ilemobade AA, Balogun TF. Pig trypanosomiasis: Effects of infection on feed intake, liveweight gain and carcass traits. Trop Anim Health Prod. 1981; 13: 128–136. <u>https://doi.org/10.1007/BF02237909</u> PMID: 7292607
- Chappuis F, Loutan L, Simarro P, Lejon V, Büscher P. Options for field diagnosis of human african trypanosomiasis. Clin Microbiol Rev. 2005; 18: 133–146. https://doi.org/10.1128/CMR.18.1.133-146.
 2005 PMID: 15653823 PMCID: PMC544181
- 58. Bisser S, Lumbala C, Nguertoum E, Kande V, Flevaud L, Vatunga G, et al. Sensitivity and Specificity of a Prototype Rapid Diagnostic Test for the Detection of *Trypanosoma brucei gambiense* Infection: A Multi-centric Prospective Study. PLoS Negl Trop Dis. 2016; 10: e0004608. https://doi.org/10.1371/ journal.pntd.0004608 PMID: 27058033 PMCID: PMC4825971
- Matovu E, Kitibwa A, Picado A, Biéler S, Bessell PR, Ndung'u JM. Serological tests for gambiense human African trypanosomiasis detect antibodies in cattle. Parasit Vectors. 2017; 10: 546. https://doi. org/10.1186/s13071-017-2487-8 PMID: 29100526 PMCID: PMC5670715
- Büscher P, Gilleman Q, Lejon V. Rapid diagnostic test for sleeping sickness. N Engl J Med. 2013; 368: 1069–1070. https://doi.org/10.1056/NEJMc1210373 PMID: 23484849
- Truc P, Lejon V, Magnus E, Jamonneau V, Nangouma A, Verloo D, et al. Evaluation of the micro-CATT, CATT/*Trypanosoma brucei gambiense*, and LATEX/*T.b. gambiense* methods for serodiagnosis and surveillance of human African trypanosomiasis in West and Central Africa. Bull World Health Organ. 2002; 80: 882–886. PMID: 12481210 PMCID: PMC2567684
- Compaoré CFA, Kaboré J, Ilboudo H, Thomas LF, Falzon LC, Bamba M, et al. Monitoring the elimination of gambiense human African trypanosomiasis in the historical focus of Batié, South–West Burkina Faso. Parasite. 2022; 29: 25. <u>https://doi.org/10.1051/parasite/2022024</u> PMID: <u>35543528</u> PMCID: PMC9093133
- **63.** Garcia A, Jamonneau V, Magnus E, Laveissière C, Lejon V, N'guessan P, et al. Follow-up of Card Agglutination Trypanosomiasis Test (CATT) positive but apparently aparasitaemic individuals in Côte d'Ivoire: evidence for a complex and heterogeneous population. Trop Med Int Health. 2000; 5: 786–793. https://doi.org/10.1046/j.1365-3156.2000.00623.x PMID: 11123826
- WHO. Control and surveillance of human African trypanosomiasis. WHO technical report series; 2013. World Health Organ Tech Rep Ser. 2013; (984):1–237. PMID: 24552089
- Fernández D, González-Baradat B, Eleizalde M, González-Marcano E, Perrone T, Mendoza M. *Trypa-nosoma evansi*: A comparison of PCR and parasitological diagnostic tests in experimentally infected mice. Exp Parasitol. 2009; 121: 1–7. https://doi.org/10.1016/j.exppara.2008.09.013 PMID: 18848544
- 66. Holland WG, Claes F, My LN, Thanh NG, Tam PT, Verloo D, et al. A comparative evaluation of parasitological tests and a PCR for *Trypanosoma evansi* diagnosis in experimentally infected water buffaloes. Vet Parasitol. 2001; 97: 23–33. https://doi.org/10.1016/S0304-4017(01)00381-8 PMID: 11337124
- Holland WG, Thanh NG, Do TT, Sangmaneedet S, Goddeeris B, Vercruysse J. Evaluation of diagnostic tests for *Trypanosoma evansi* in experimentally infected pigs and subsequent use in field surveys in north Vietnam and Thailand. Trop Anim Health Prod. 2005; 37: 457–467. <u>https://doi.org/10.1007/</u> s11250-005-1217-y PMID: 16248217
- Monzón CM, Mancebo OA, Roux JP. Comparison between six parasitological methods for diagnosis of *Trypanosoma evansi* in the subtropical area of Argentina. Vet Parasitol. 1990; 36: 141–146. https://doi. org/10.1016/0304-4017(90)90102-h PMID: 2382382
- Paris J, Murray M, McOdimba F. A comparative evaluation of the parasitological techniques currently available for the diagnosis of African trypanosomiasis in cattle. Acta Trop. 1982; 39:307–316. PMID: 6131590
- 70. Pruvot M, Kamyingkird K, Desquesnes M, Sarataphan N, Jittapalapong S. A comparison of six primer sets for detection of *Trypanosoma evansi* by polymerase chain reaction in rodents and Thai livestock. Vet Parasitol. 2010; 171: 185–193. https://doi.org/10.1016/j.vetpar.2010.04.001 PMID: 20466489
- Sharma P, Juyal PD, Singla LD, Chachra D, Pawar H. Comparative evaluation of real time PCR assay with conventional parasitological techniques for diagnosis of *Trypanosoma evansi* in cattle and buffaloes. Vet Parasitol. 2012; 190: 375–382. https://doi.org/10.1016/j.vetpar.2012.07.005 PMID: 22858228
- Hassan-Kadle AA, Ibrahim AM, Nyingilili HS, Yusuf AA, Vieira RFC. Parasitological and molecular detection of *Trypanosoma* spp. in cattle, goats and sheep in Somalia. Parasitology. 2020; 147: 1786– 1791. https://doi.org/10.1017/S003118202000178X PMID: 32951618
- 73. Muleed MA, Chaudhri M, Shakoori A. Comparative studies on the sensitivity of polymerase chain reaction (PCR) and microscopic examination for the detection of *Trypanosoma evansi* in horses. Turk J Vet Anim Sci. 2010; 34: 507–512. https://doi.org/10.3906/vet-0806-22 PMID: 9681244

- 74. Ebhodaghe F, Isaac C, Ohiolei JA. A meta-analysis of the prevalence of bovine trypanosomiasis in some African countries from 2000 to 2018. Prev Vet Med. 2018; 160: 35–46. https://doi.org/10.1016/j. prevetmed.2018.09.018 PMID: 30388996
- 75. Sloof P, Bos JL, Konings AFJM, Menke HH, Borst P, Gutteridge WE, et al. Characterization of satellite DNA in *Trypanosoma brucei* and *Trypanosoma cruzi*. J Mol Biol. 1983; 167: 1–21. <u>https://doi.org/10.1016/S0022-2836(83)80031-X PMID: 6345792</u>
- 76. Gonzalez A, Prediger E, Huecas ME, Nogueira N, Lizardi PM. Minichromosomal repetitive DNA in *Trypanosoma cruzi*: its use in a high-sensitivity parasite detection assay. Proc Natl Acad Sci U S A. 1984; 81: 3356–3360. https://doi.org/10.1073/pnas.81.11.3356 PMID: 6427769 PMCID: PMC345506
- 77. Desquesnes M, Dávila AMR. Applications of PCR-based tools for detection and identification of animal trypanosomes: a review and perspectives. Vet Parasitol. 2002; 109: 213–231. https://doi.org/10.1016/s0304-4017(02)00270-4 PMID: 12423934
- 78. Suprihati E, Suwanti LT, Yudhana A, Kusumaningrum AI. Comparison of ITS-1 and TBR-1/2 primer sensitivity for the detection of *Trypanosoma evansi* local isolates in experimental rats using a polymerase chain reaction. Vet World. 2022; 1772–1778. https://doi.org/10.14202/vetworld.2022.1772–1778 PMID: 36185509 PMCID: PMC9394150
- 79. Ahmed HA, Picozzi K, Welburn SC, MacLeod ET. A comparative evaluation of PCR- based methods for species- specific determination of African animal trypanosomes in Ugandan cattle. Parasit Vectors. 2013; 6: 316. https://doi.org/10.1186/1756-3305-6-316 PMCID: PMC4029050 PMID: 24499678
- 80. Van Reet N, Patient Pyana P, Dehou S, Bebronne N, Deborggraeve S, Büscher P. Single nucleotide polymorphisms and copy-number variations in the Trypanosoma brucei repeat (TBR) sequence can be used to enhance amplification and genotyping of Trypanozoon strains. PloS One. 2021; 16: e0258711. https://doi.org/10.1371/journal.pone.0258711 PMID: 34695154 PMCID: PMC8544829
- **81.** Dama E, Camara O, Kaba D, Koffi M, Camara M, Compaoré C, et al. Immune trypanolysis test as a promising bioassay to monitor the elimination of *gambiense* human African trypanosomiasis. Parasite. 2019; 26: 68. https://doi.org/10.1051/parasite/2019066 PMID: 31755862 PMCID: PMC6873819
- 82. Desquesnes M. Livestock trypanosomoses and their vectors in Latin America. Paris: World Organization for Animal Health (OIE). 2004; 174 p.
- Desquesnes M, Bossard G, Thévenon S, Patrel D, Ravel S, Pavlovic D, et al. Development and application of an antibody-ELISA to follow up a *Trypanosoma evansi* outbreak in a dromedary camel herd in France. Vet Parasitol. 2009; 162: 214–220. https://doi.org/10.1016/j.vetpar.2009.03.033 PMID: 19372008
- Pruvot M, Kamyingkird K, Desquesnes M, Sarataphan N, Jittapalapong S. The effect of the DNA preparation method on the sensitivity of PCR for the detection of *Trypanosoma evansi* in rodents and implications for epidemiological surveillance efforts. Vet Parasitol. 2013; 191: 203–208. <u>https://doi.org/10.1016/j.vetpar.2012.09.010 PMID: 23040401</u>
- OIE. Principles and methods of validation of diagnostic assays for infectious disease. Chapter 1.1.6. Terrestrial Manual. 2018; pp. 72–87
- Büscher P. Diagnosis of african trypanosomiasis. Magez S., Radwanska M. (eds.). Trypanosomes and trypanosomiasis. Magez S., Radwanska M. (eds.). springer, wien, 2014. pp. 189–216.
- Desquesnes M, Bengaly Z, Dia ML. Évaluation de la persistance des anticorps détectés par Elisa-indirect *Trypanosoma vivax* après traitement trypanocide chez des bovins naturellement infectés. Revue Élev. Méd. vét. Pays trop. 2003; 56:141–144. https://doi.org/10.19182/remvt.9855
- Van Meirvenne N, Magnus E, Buscher P. Evaluation of variant specific trypanolysis tests for serodiagnosis of human infections with *Trypanosoma brucei gambiense*. Acta Trop. 1995; 60: 189–199. <u>https:// doi.org/10.1016/0001-706x(95)00127-z PMID: 8907397</u>
- Desquesnes M, Sazmand A, Gonzatti M, Boulangé A, Bossard G, Thévenon S, et al. Diagnosis of animal trypanosomoses: proper use of current tools and future prospects. Parasit Vectors. 2022; 15: 235. https://doi.org/10.1186/s13071-022-05352-1 PMID: 35761373 PMCID: PMC8858479